Author's response to reviews

Title: Safety, feasibility and effects of an individualised walking intervention for women undergoing chemotherapy for ovarian cancer: a pilot study.

Authors:

Melissa J Newton (mj.newton@qut.edu.au)
Sandi C Hayes (sc.hayes@qut.edu.au)
Monika Janda (m.janda@qut.edu.au)
Penelope M Webb (penny.webb@qimr.edu.au)
Andreas Obermair (obermair@powerup.com.au)
Elizabeth G Eakin (e.eakin@sph.uq.edu.au)
David Wyld (David_Wyld@health.qld.gov.au)
Louisa G Gordon (louisa.gordon@griffith.edu.au)
Vanessa L Beesley (Vanessa.Beesley@qimr.edu.au)

Version: 9 Date: 2 September 2011

Author's response to reviews:

Changed the format of Figure 2.